Author's response to reviews

Title: Intensified specimen collection to improve tuberculosis diagnosis in children from rural South Africa, an observational study.

Authors:

Tania A Thomas (tthomasheysell@gmail.com)
Scott K Heysell (skh8r@virginia.edu)
Prashini Moodley (moodleyp@ukzn.ac.za)
Romualde Montreuil (romy1115@hotmail.com)
Xia Ha (xha@montefiore.org)
Gerald Friedland (gerald.friedland@yale.edu)
Sheila A Bamber (sabamber@telkosma.net)
Anthony P Moll (tonymoll@telkosma.net)
Neel Gandhi (neelgandhi@alumni.williams.edu)
William E Brant (WEB9C@hscmail.mcc.virginia.edu)
A Willem Sturm (sturm1@ukzn.ac.za)
N Sarita Shah (bwg2@cdc.gov)

Version: 3
Date: 20 December 2013

Author's response to reviews:

Please see attached letter with responses to reviewers.